Health and Quality of Life Outcomes

Research

The validity and responsiveness of three quality of life measures in the assessment of psoriasis patients: results of a phase II study Richard Shikiar¹, Mary Kaye Willian², Martin M Okun³, Christine S Thompson¹ and Dennis A Revicki^{*1}

Address: ¹Center for Health Outcomes Research, United BioSource Corporation, Bethesda, MD, USA, ²Global Health Economics and Outcomes Research, Abbott Laboratories, Abbott Park, IL, USA and ³Abbott Immunology, Abbott Laboratories, Abbott Park, IL, USA

Email: Richard Shikiar - rich.shikiar@unitedbiosource.com; Mary Kaye Willian - mary.willian@abbott.com; Martin M Okun - martin.okun@abbott.com; Christine S Thompson - chris.thompson@unitedbiosource.com; Dennis A Revicki* - dennis.revicki@unitedbiosource.com

* Corresponding author

Published: 27 September 2006

Health and Quality of Life Outcomes 2006, 4:71 doi:10.1186/1477-7525-4-71

This article is available from: http://www.hqlo.com/content/4/1/71

© 2006 Shikiar et al; licensee BioMed Central Ltd.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (<u>http://creativecommons.org/licenses/by/2.0</u>), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Received: 24 August 2006 Accepted: 27 September 2006

Abstract

Background: Patient-reported outcome (PROs) measures are being used more frequently in investigational studies of treatments for moderate to severe plaque psoriasis. The objective of this study was to examine the relationships among the Dermatology Life Quality Index (DLQI), the Short Form 36 (SF-36), and the EuroQOL 5D (EQ-5D) and to assess their validity, responsiveness, and estimates of minimum important differences.

Methods: A Phase II, randomized, double-blind, parallel group, placebo-controlled, multi-center clinical trial assessed the clinical efficacy and safety of two doses of subcutaneously administered adalimumab vs. placebo for 12 weeks in the treatment of 147 patients with moderate to severe plaque psoriasis. This study provided the opportunity to evaluate the validity and responsiveness to change in clinical status of PROs instruments. Patients completed the DLQI, SF-36, and EQ-5D questionnaires at baseline and at 12 weeks. Blinded investigators assessed the Psoriasis Area and Severity Index (PASI) scores and the Physician's Global Assessment (PGA) scores of enrolled patients. The responsiveness of the measures to changes in the clinical endpoints from baseline to Week 12 was assessed. Estimates of minimum important differences (MID) were derived. All analyses were performed with blinded data; findings and conclusions were not biased based on treatment condition.

Results: The dermatology-specific DLQI was highly correlated to clinical endpoints at baseline and at Week 12, and was the most responsive PRO to changes in endpoints. Compared with the SF-36, the EQ-5D index score and VAS scores were generally more highly correlated with clinical endpoints, but displayed about the same degree of responsiveness. The most responsive SF-36 scales were the Bodily Pain and Social Functioning scales. Estimates of the MID for the DLQI ranged from 2.3–5.7 and for the SF-36 Physical Component Summary (PCS) score ranged from 2.5–3.9.

Conclusion: This study provides support for the continued use of the DLQI and SF-36 PCS in the assessment of treatments for psoriasis. On the basis of the results from this trial, the EQ-5D should be considered as a general PRO measure in future clinical trials of patients with moderate to severe plaque psoriasis.

Open Access

Background

Moderate to severe plaque psoriasis has been demonstrated to have substantial impact on function limitations and psychosocial factors of patients with the disease [1-5]. Moreover, successful treatment of moderate to severe psoriasis – as assessed by improved physical functioning and reduction of signs and symptoms – has been shown to have a positive impact on social and psychological aspects of psoriasis [6-11].

Given the functional and psychosocial impact of the disease, studies of moderate to severe psoriasis patients often include both physician-assessed clinical endpoints and dermatology-specific patient-reported outcomes (PROs) to obtain a holistic view of the disease and treatment effects in patients [12]. Such practices are bolstered by the assertion of the Medical Advisory Board of the National Psoriasis Foundation (NPF) that, even more so than physical signs, such as the percentage of body surface area (BSA) affected by psoriasis, the severity of psoriasis is "first and foremost a quality-of-life (QOL) issue" [13]. The same values for percentage BSA involvement can result in very different degrees of impact for different patients, depending on the location of psoriatic plaques, the pain associated with the lesions and plaques, the extent of bleeding associated with the psoriatic lesions, and resulting functional limitations. The NPF Advisory Board suggests an alternative basis for defining mild, moderate, or severe psoriasis, predicated on QOL impacts of the disease. Similarly, the guidelines recently promulgated by the British Association of Dermatologists [14] for the use of biologics in psoriasis indicate that eligible patients must have a Psoriasis Area and Severity Index (PASI) score of at least 10 and a score on the Dermatology Life Quality Index (DLQI) [15] - a dermatology-specific validated PRO measure - of greater than 10.

A Phase II clinical trial of two dosages of adalimumab and placebo in the treatment of moderate to severe psoriasis provided an opportunity to further explore the psychometric characteristics - including responsiveness and minimum important differences - of the three PROs used in the trial: the DLQI; the general health-related QOL measure MOS Short Form 36 (SF-36) Health Survey [16]; and the general health status measure EuroQOL 5D (EQ-5D) [17,18]. Establishing the reliability, validity, and responsiveness of PRO measures is necessary for their use in support of labeling claims, according to an FDA draft guide to industry [19]. Reliability refers to the accuracy of a measure, while validity refers to the extent the measure actually is measuring what it purports to measure. Responsiveness is a component of validity and represents the PRO's capability to detect changes related to changes in the clinical status of patients or other relevant outcomes measures. Minimum important difference (MID) is related to

responsiveness and provides guidance to those reviewing clinical trial results as to whether the statistically significant group differences or changes are clinically meaningful and important. Jaeschke and colleagues [20] define a minimal clinically important difference (MCID) (we use MID instead of MCID to avoid confusion) as "the smallest difference in score ... which patients perceive as beneficial and which would mandate, in the absence of troublesome side-effects and excessive cost, a change in the patient's management." Estimation of the MID – using several different approaches – is also emphasized in the FDA guidance and is consistent with recently published recommendations of health outcomes researchers [21,22].

Methods

Overview

The objectives of the Phase II, randomized, double-blind, parallel group, placebo-controlled, multi-center clinical trial were to assess the clinical efficacy and safety of subcutaneously administered adalimumab vs. placebo using two dosage regimens for 12 weeks in the treatment of 147 patients with moderate to severe plaque psoriasis. The study included a screening period, a blinded 12-week treatment period, and a 30-day follow-up visit for patients not completing 12 weeks of active treatment or not entering an extension study. Time between screening and baseline visits was not to exceed 28 days. The trial achieved the objectives of the study in terms of safety and clinical efficacy endpoints [23].

Patients and inclusion criteria

Patients with a diagnosis of moderate to severe plaque psoriasis and an affected BSA of \geq 5% for at least 1 year were eligible for the study. In addition to other inclusion criteria (e.g., age \geq 18 years, willingness to give informed consent), patients had to be able to self-inject medication or have a designee or nurse who could inject the randomized assignment. Patients signed informed consent forms, and the study complied with FDA Good Clinical Practices, Health Protection Branch guidelines, and all other applicable ethical, legal, and regulatory requirements [23].

Clinical measures

For purposes of the analyses reported here, there were two primary clinical outcomes:

Psoriasis Area and Severity Index

Frequently used as an endpoint in psoriasis clinical trials [24], the PASI [25] was the primary efficacy outcome in this trial. PASI is a composite index indicating the severity of the three main signs of psoriatic plaques (i.e., ery-thema, scaling, and thickness) and is weighted by the amount of coverage of these plaques in the four main

body areas (head, trunk, upper extremities, and lower extremities). PASI scores range from 0–72, with higher scores indicating greater disease severity. PASI was assessed at screening and baseline, at Weeks 1, 2, 4, 8, and 12/Early Termination, and at the final follow-up visit.

Physician's Global Assessment

The PGA is a seven-point scale used to measure the severity of disease at the time of the physician's evaluation. The seven disease categories are:

- *Severe*: very marked plaque elevation, scaling, and/or erythema
- *Moderate to Severe*: marked plaque elevation, scaling, and/or erythema
- *Moderate*: moderate plaque elevation, scaling, and/or erythema
- *Mild to moderate*: intermediate between moderate and mild
- Mild: slight plaque elevation, scaling, and/or erythema
- Almost Clear: intermediate between mild and clear

• *Clear*: no signs of psoriasis (post-inflammatory hypopigmentation or hyperpigmentation could be present).

The PGA scale is scored from 1 (Clear) to 7 (Severe). The PGA was assessed by the investigator at screening, baseline, and Weeks 1, 2, 4, 8, 12/Early Termination, and the follow-up visit. Each study site was to make every attempt to have the same investigator perform these assessments throughout the study for each patient.

Patient-Reported Outcome measures

Three PROs measures were used in the study and are the subject of the analyses reported here. All PROs measures were assessed at baseline and at Week 12 (or early termination, if applicable).

Dermatology Life Quality Index

The DLQI was developed as a simple, compact, and practical questionnaire for use in dermatology clinical settings to assess limitations related to the impact of skin disease [15]. The instrument contains 10 items dealing with skin (e.g., Item 1: "Over the last week, how itchy sore, painful, or stinging has your skin been?"). The DLQI score ranges from 0–30, with "30" corresponding to the worst quality of life, and "0" corresponding to the best score. The DLQI has well-established properties of reliability and validity in the dermatology setting [15,26-28].

Short Form 36 health survey

The SF-36 is a 36-item general health status instrument often used in clinical trials and health services research [16]. It consists of eight domains: Physical Function, Role Limitations-Physical, Vitality, General Health Perceptions, Bodily Pain, Social Function, Role Limitations -Emotional, and Mental Health. Two overall summary scores can be obtained - a Physical Component Summary (PCS) score and a Mental Component Summary (MCS) score [29]. The PCS and MCS scores range from 0-100, with higher scores indicating better health. The SF-36 has been used in a wide variety of studies involving psoriasis, including descriptive studies [30] and clinical research studies [6,7], and has demonstrated good reliability and validity. Internal consistency for most SF-36 domains is greater than 0.70. The SF-36 has been shown to discriminate between known groups in a variety of diseases, is reproducible, and is responsive to longitudinal clinical changes.

EuroQOL 5D

The EQ-5D [17,18] is a six-item, preference-based instrument designed to measure general health status. The EQ-5D has two sections: The first consists of five items to assess degree of physical functioning (mobility, self-care, usual activities, pain/discomfort, and anxiety/depression). Items are rated on a three-point scale ranging from "No Problem" to "Extreme Problem" or "Unable to Do." Each pattern of scores for the five items is linked to an index score that has a value ranging from 0-1, indicating the health utility of that person's health status. The specific linkage can differ from country to country, reflecting differences in cultures to the item responses. The second section is the sixth item on the EQ-5D, which is a visual analog scale with endpoints of "100" or "Best Imaginable Health," and "0" or "Worst Imaginable Health." It offers a simple method for the respondents to indicate how good or bad their health statuses are "today." The score is taken directly from the patients' responses.

Statistical methods

Validity of the PRO measures was assessed in several ways. First, an assessment was made of the concurrent validity of scales and subscales (i.e., the extent to which PRO measures are correlated with one another). As a disease-specific PRO measure, the DLQI was expected to correlate moderately to extremely well with general PRO measures. Another important aspect of validity in this study was to assess the extent to which the PRO measures correlated with the clinical endpoints – PASI and PGA – both at baseline and at Week 12.

Responsiveness of PRO measures was assessed via two approaches. First, changes in these measures from baseline to Week 12 were correlated with changes in the PASI or PGA over the 12-week course of treatment within the trial. Concurrent improvement in both clinical measures and PRO measures was expected to result in positive correlations. The second approach to assessing responsiveness involved categorizing patients into responder groups based on the changes in their PASI scores from baseline to Week 12. This was done in two ways. First, a responder was defined as a patient with >75% improvement in PASI (consistent with the definition of success with the primary efficacy variable), and a non-responder was defined as a patient with a PASI improvement <50% (consistent with the definition of failure for a secondary efficacy variable). Tests of mean differences in improvement on the PRO measures were completed between the two groups. Secondly, in support of the estimation of the MID, discussed below, patients were further categorized by degree of PASI response, and assessed differences among these four groups: PASI improvement <25%; PASI improvement 25-49%; PASI improvement 50-74%; and PASI improvement \geq 75%. Analyses of variance tests were performed among these four groups for changes in PRO measures.

In accordance with the FDA draft guidance [19] and consistent with recent recommendations from PRO researchers [21,22], five methods were used to estimate MIDs of the PROs. The PRO change score corresponding to PASI 25-PASI 49 was the first estimate of MID, called MID-1. This was based on the assumption that patients would perceive a PASI improvement of 25% as beneficial. The trial did not provide data to test this assumption (e.g., there was no rating by patients of their overall improvements). A second estimate, MID-2, was based on the PRO change score corresponding to a PASI improvement between 50-74%. The PASI 50 is seen as clinically relevant, and, as such, this degree of improvement served as a secondary efficacy endpoint in this trial. A third method for estimating MID relied on the association of changes in the PRO measure with changes in the PGA. A nonresponder was defined as a patient with a PGA change score of either "0" (no change) or "1" (slight increase in severity of disease) from baseline to Week 12. A minimal responder was defined as a patient whose PGA improved by either 1 or 2 points from baseline to Week 12. The third estimate of MID, MID-3, was the difference in the PRO score between non-responders and minimal responders.

In addition, two distributional methods were used to support the anchor-based MID estimates for the PROs [21,22]. Based on evidence by Wyrwich and associates [31,32], the standard error of measurement (SEM) can be used to approximate the MID. The SEM, which describes the error associated with the measure, was estimated by the standard deviation of the measure multiplied by the square root of 1 minus its reliability coefficient. Finally, there has been discussion [33] concerning a number of

studies demonstrating that one-half of the standard deviation of a measure represents the upper limit of the MID [22]. In estimating the SEM for the SF-36 and the EQ-5D, reliability estimates from the literature were used. The SEM for the DLQI incorporated the reliability estimated from the trial data, which was consistent with what has been found in the literature for this instrument [27].

Finally, it is important to note that all analyses were performed with blinded data (i.e., the statuses of patients with respect to their assigned treatment groups were not known).

Results

Patient demographics and clinical characteristics

A total of 147 patients enrolled and received at least one dose of study medication at 18 sites in the United States and Canada. Blinded data were available for the PROs for 147 patients at baseline and 140 patients at Week 12. Since the focus of these analyses were on the psychometric properties of the PROs rather than with efficacy, observed cases were employed rather than last observation carried forward or other methods for treating missing observations at the end of trial. The mean age of the patients enrolled in the trial was 44.2 years, two-thirds were male, and the preponderance were white (Table 1).

Clinical endpoints

The results for the PASI and the PGA at baseline and Week 12, as well as the change from baseline to Week 12, are displayed in Table 2. The mean PASI at baseline was 15.7, which decreased by 8.9 points (improvement) to 6.8 by Week 12. The mean PGA at baseline was 5.5 (i.e., midway between "Moderate" and "Moderate to Severe"), and decreased (improved) by 2.1 points to 3.4 by Week 12 (i.e., between "Mild" and "Mild to Moderate"). In evaluating the improvement in the two clinical endpoints, it is important to keep in mind that these analyses included pooled placebo and active treatment groups.

Characteristic	(N = 147)
Age	
Mean (SD)	44.2 (12.7)
Gender	
Female n (%)	48 (32.7%)
Male n (%)	99 (67.3%)
Race	
White n (%)	133 (90.5%)
Black n (%)	4 (2.7%)
Asian n (%)	5 (3.4%)
Other n (%)	5 (3.4%)

Measure	Baseline	Week I2	Change ²
PASI	15.69 (7.34)	6.84 (7.77)	-8.87 (8.41)
PGAI	5.48 (0.81)	3.36 (1.74)	-2.14 (1.87)

^I Scored such that I = "Clear" to 7 = "Severe."

²Change scores are computed only for the 140 patients with scores at baseline and Week 12; sample size at baseline = 147.

Patient-Reported Outcome measures

The results for the DLQI, SF-36, and EQ-5D at baseline and Week 12, and the change from baseline are shown in Table 3. Based on blinded data, mean PRO measures improved during the course of the trial (a decrease in DLQI scores indicates an improvement; an increase in the SF-36 and EQ-5D indicates improvement). The greatest improvement in a DLQI item occurred for the first item, assessing how "itchy, sore, painful, or stinging" the person's skin felt (data not shown). Similarly, as shown in Table 3, the greatest improvement among the SF-36 scales was for Bodily Pain. The largest improvement among the five EQ-5D dimensions occurred for the Pain/Discomfort dimension (data not shown). Given these findings, it appears that improvement in pain and discomfort is the most pronounced among all PRO measures assessed.

The reliability of the DLQI, as assessed by coefficient alpha, was 0.89 at baseline and 0.92 at Week 12, indicating that this is a highly reliable measure, and in line with previous findings [27,28].

Relationships among Patient-Reported Outcome measures

Table 4 displays the correlations among PRO measures at baseline and at Week 12, as well as the correlations among

changes in these measures from baseline to Week 12. There were a few trends evident form this data. First, all measures were statistically significantly inter-correlated. Second, with respect to the relationship between the DLQI and the SF-36, the DLQI correlated the greatest with the Bodily Pain and Social Functioning domains, both at baseline and at Week 12, and, for changes in these scores over the course of the trial. Third, the DLQI correlated highly with the EQ-5D index score, and these correlations were consistently higher than the correlations with the EQ-5D visual analog scale (VAS) scores. Fourth, the EQ-5D index score tended to correlate greatest with the Bodily Pain domain of the SF-36. Finally, the scores tended to be more highly correlated at the end of the trial than at baseline, consistent with previous findings [28].

Correlations with clinical endpoints

Table 5 displays correlations of PRO measures with the two clinical assessments – PASI score and PGA – at baseline (first two columns of data) and at Week 12 (second two columns). In addition to almost uniformly greater correlations at Week 12 vs. at baseline – consistent with previous findings [28] – one can also note that both the DLQI and EQ-5D index score tended to be more highly correlated with the two clinical endpoints than any of the SF-36 domains. The SF-36 scales with the strongest associ-

Table 3: Mean (Standard Deviation	 of PROs at baseline and week 12, and chan 	ge from baseline to week 12

leasure	Baseline	Week 12	Change
DLQI Total Score	12.71 (7.18)	5.28 (6.49)	-7.45 (7.78)
F-36			
Physical Functioning	79.26 (24.95)	84.43 (22.62)	5.63 (22.25)
Role-Physical	72.45 (37.56)	82.50 (34.67)	9.64 (40.30)
Bodily Pain	59.58 (25.37)	75.87 (24.89)	16.59 (26.90)
General Health	66.7 (20.6)	72.69 (20.79)	6.10 (16.25)
Vitality	53.89 (22.81)	61.04 (23.18)	7.60 (20.88)
Social Functioning	74.49 (27.56)	85.98 (23.63)	11.16 (27.80)
Role-Emotional	76.03 (35.39)	85.48 (30.77)	8.39 (37.02)
Mental Health	69.39 (19.30)	77.43 (17.73)	8.14 (18.57)
Physical Summary Score (PCS)	47.93 (10.23)	51.24 (9.51)	3.47 (9.30)
Mental Summary Score (MCS)	47.30 (11.23)	51.36 (10.08)	3.94 (11.04)
Q-5D			
Index Score	0.66 (0.28)	0.82 (0.23)	0.16 (0.29)
VAS Overall Health	72.25 (20.67)	81.22 (17.26)	9.35 (20.71)

¹Change scores are computed only for patients with scores at baseline and Week 12; this number varied between 138 and 140, depending on the specific measure, as compared with the 147 patients at baseline.

Measure		Baseline			Week 12			Change	
	DLQI Total	EQ-5D Index	EQ-5D VAS	DLQI Total	EQ-5D Index	EQ-5D VAS	DLQI Total	EQ-5D Index	EQ-5D VAS
DLQI Total Score SF-36	1.00	-0.51	-0.35	1.00	-0.71	-0.58	1.00	-0.53	-0.46
Physical Functioni ng	-0.44	0.58	0.35	-0.41	0.61	0.49	-0.29	0.47	0.32
Role- Physical	-0.45	0.64	0.38	-0.57	0.67	0.50	-0.47	0.51	0.45
Bodily Pain	-0.55	0.73	0.45	-0.61	0.76	0.56	-0.66	0.64	0.53
General Health	-0.24**	0.39	0.47	-0.38	0.59	0.69	-0.33	0.46	0.46
Vitality	-0.31	0.43	0.48	-0.43	0.62	0.60	-0.46	0.37	0.48
Social Functioni ng	-0.69	0.52	0.46	-0.68	0.74	0.60	-0.68	0.50	0.56
Role- Emotiona I	-0.41	0.45	0.42	-0.56	0.67	0.53	-0.50	0.41	0.48
Mental Health	-0.44	0.46	0.50	-0.56	0.66	0.67	-0.52	0.49	0.56
Physical Summary Score (PCS)	-0.41	0.64	0.36	-0.46	0.65	0.52	-0.41	0.56	0.39
Mental Summary Score (MCS) EQ-5D	-0.45	0.39	0.49	-0.58	0.66	0.63	-0.59	0.42	0.57
Index Score	-0.51	1.00	0.39	-0.71	1.00	0.63	-0.53	1.00	0.39
VAS- General Health	-0.35	0.39	1.00	-0.58	0.63	1.00	-0.46	0.39	1.00

Table 4: Correlations¹ among PROs at baseline and week 12, and change from baseline to week 12

¹All correlations were significant at p < 0.001, unless otherwise noted. * $p \le 0.05$, ** $p \le 0.01$, ns = non-significant.

ation with clinical endpoints are Social Functioning and Bodily Pain.

Responsiveness of the Patient-Reported Outcome measures

An important attribute for a PRO measure is responsiveness to change in the clinical status of a patient (i.e., as a patient's disease improves, the PRO measures also improve). The last two columns of Table 5 display the correlations between changes in PRO measures used in the trial and changes in PASI scores and the PGA from baseline to Week 12. These data demonstrate that the DLQI is the most responsive of the PRO measures. The correlations between changes over the course of the trial in the DLQI total score and changes in the PASI score (r = 0.69, p < 0.001) and PGA (r = 0.71, p < 0.001) approach the correlation between changes in the two clinical measures themselves (r = 0.75, p < 0.001). In addition, the DLQI is the only one of the PRO measures to demonstrate equal responsiveness to PGA and PASI scores. The correlation between changes in the EQ-5D index score and the two clinical assessments was r = -0.57 (p < 0.001) for changes in the PASI to r = -0.44 for changes in the PGA (p < 0.001). Similarly, the correlations between changes in all but one of the SF-36 scores and changes in the PGA were smaller than correlations between changes in the SF-36 and the PASI.

A second way to assess responsiveness was to contrast patients who were defined as clinical responders with those characterized as non-responders. Given that the primary endpoint in the trial was defined as the percentage of patients achieving a PASI 75 response (i.e., $\geq 75\%$ improvement in PASI from. baseline) by Week 12, a

Measure	Base	eline	We	ek I2	Cł	Change		
	PASI	PGA	PASI	PGA	PASI	PGA		
DLQI Total Score	0.31	0.29	0.67	0.65	0.69	0.71		
SF-36								
Physical Functioning	-0.32	-0.14 ns	-0.28	-0.25***	-0.38	-0.14 ns		
Role-Physical	-0.22**	-0.14 ns	-0.41	-0.37	-0.42	-0.31		
Bodily Pain	-0.36	-0.19*	-0.47	-0.42	-0.60	-0.44		
General Health	-0.08 ns	0.05 ns	-0.34	-0.33	-0.34	-0.24**		
Vitality	-0.15 ns	-0.06 ns	-0.37	-0.37	-0.38	-0.31		
Social Functioning	-0.23**	-0.21**	-0.46	-0.38	-0.44	-0.43		
Role-Emotional	-0.16 ns	-0.06 ns	-0.37	-0.29	-0.39	-0.36		
Mental Health	-0.17*	-0.09 ns	-0.46	-0.38	-0.43	-0.36		
Physical Summary	-0.28	-0.13 ns	-0.35	-0.33	-0.45	-0.25**		
Mental Summary	-0.12 ns	-0.08 ns	-0.44	-0.36	-0.40	-0.42		
EQ-5D								
Index Score	-0.40	-0.31	-0.60	-0.51	-0.57	-0.44		
VAS-General Health	-0.24**	-0.09 ns	-0.52	-0.43	-0.43	-0.38		
PASI	1.00	0.59	1.00	0.83	1.00	0.75		

Table 5: Correlations¹ between PROs and clinical endpoints at baseline and week 12, and change from baseline to week 12

¹All correlations were significant at p < 0.001, unless otherwise noted. * $p \le 0.05$, ** p < 0.01, ns = non-significant.

responder was defined as a patient with a PASI75 response. A non-responder was a patient with <PASI 50, since some of the secondary endpoints in the trial used this cut-off. The results of these analyses are displayed in Table 6. DLQI total scores for responders improved by 12.17 points, while scores of non-responders improved by 1.77 points. This difference was statistically significant (t = 9.0; p < 0.0001). All the PRO measures except for the SF-36 Physical Functioning domain were responsive, as defined by a statistically significant difference between responders and non-responders. The DLQI was the most responsive of the PRO measures, as evidenced by the size of the t-statistic and the effect size. The responsiveness of

the EQ-5D index and VAS scores were generally the same as several of the SF-36 domain scores.

While the estimates of responsiveness displayed in the last two columns of Table 5 take into account the full range of PASI change scores and their relationship to PRO change scores, the responsiveness analysis in Table 6 places patients in two categories – responders and non-responders. Table 7 defines *four* categories of responders: responders, defined as those with PASI improvements \geq 75%; "partial responders," those with PASI improvement 50– 74%, inclusively; "near responders," those with PASI improvement 25–49%, inclusively; and non-responders,

Table 6: Change in PRC) measures among	responder ¹ groups
------------------------	------------------	-------------------------------

Change in Measure	Mean Change Score for Responders (n = 66)	Mean Change Score for Non-Responders (n = 53)	Difference	t-value	P Value	Effect Size
DLQI Total Score	-12.17 (6.78)	-1.77 (5.52)	-10.39	9.0	<.0001	0.40
SF-36						
Physical Functioning	9.12 (23.50)	3.52 (20.19)	5.59	1.4	0.1724	0.01
Role-Physical	20.08 (35.69)	-5.19 (44.76)	25.26	3.4	0.0008	0.08
Bodily Pain	26.47 (27.40)	4.21 (22.74)	22.26	4.7	<.0001	0.15
General Health	8.87 (15.62)	1.47 (17.77)	7.39	2.4	0.0178	0.04
Vitality	13.01 (22.58)	1.13 (18.20)	11.87	3.1	0.0024	0.07
Social Functioning	21.59 (28.13)	-2.59 (25.04)	24.19	4.9	<.0001	0.16
Role-Emotional	19.70 (32.54)	-9.62 (36.95)	29.31	4.6	<.0001	0.14
Mental Health	14.55 (17.77)	-0.38 (18.15)	14.92	4.5	<.0001	0.14
Physical Summary Score (PCS)	5.35 (9.67)	1.47 (9.08)	3.88	2.2	0.0287	0.03
Mental Summary Score (MCS)	8.03 (10.59)	-2.03 (10.42)	10.06	5.1	<.0001	0.17
EQ-5D						
Index Score	0.25 (0.30)	0.04 (0.26)	0.22	4.2	<.0001	0.12
VAS General Health	15.69 (18.96)	1.92 (23.24)	13.77	3.5	0.0006	0.09

¹Responder is defined as PASI improvement \geq 75%; non-responder is defined as PASI improvement <50%.

Change in Measure	PASI Improvement <25% (n = 3 l)	PASI Improvement 25–49% (n = 22)	PASI Improvement 50–74% (n = 21)	PASI Improvement ≥75% (n = 66)	Overall F Value	p Values ⁱ
DLQI Total Score	-0.16 (5.41)	-4.05 (4.95)	-6.95 (5.71)	-12.17 (6.78)	30.4***	2**,3***,5***,6*
SF-36						
Physical Functioning	2.15 (17.67)	5.45 (23.60)	0.02 (22.36)	9.12 (23.50)	1.2	
Role-Physical	-11.29 (39.18)	3.41 (51.35)	14.29 (31.20)	20.08 (35.69)	4.9**	3**
Bodily Pain	-1.03 (21.42)	11.59 (22.96)	16.76 (22.74)	26.47 (27.40)	9.0***	3***
General Health	-0.61 (17.33)	4.41 (18.37)	9.19 (11.29)	8.87 (15.62)	2.8*	
Vitality	-1.45 (16.29)	4.77 (20.44)	6.90 (17.43)	13.01 (22.58)	3.8*	3*
Social	-3.23 (23.71)	-1.70 (27.36)	13.10 (17.44)	21.59 (28.13)	8.7***	3***, 5**
Functioning						
Role-Emotional	-11.11 (36.44)	-7.58 (38.40)	17.46 (34.35)	19.70 (32.54)	7.6 ***	2*, 3**, 5*
Mental Health	-0.77 (18.08)	0.18 (18.66)	9.52 (13.53)	14.55 (17.77)	7.2***	3**,5*
Physical Summary	-0.31 (7.18)	3.91 (10.88)	2.57 (7.78)	5.35 (9.67)	2.7*	
Mental Summary	-2.19 (9.86)	-1.82 (11.38)	6.05 (6.90)	8.03 (10.59)	9.9 ^{****}	2*, 3***, 5**
EQ-5D						
Index Score	-0.01 (0.26)	0.10 (0.24)	0.20 (0.21)	0.25 (0.30)	7.1 ****	3***
VAS General Health	0.58 (24.31)	3.82 (22.07)	8.43 (11.24)	15.69 (18.96)	4.8 **	3**
Mean PASI Improvement ²	0.94 (4.07)	-6.24 (2.99)	-8.94 (3.47)	-14.33 (7.65)		

Table 7: PRO change scores corresponding to levels of PASI improvement

¹Pairwise comparisons between means were performed using Scheffe's test adjusting for multiple comparisons. I = improvement <25% vs. improvement 25–49%, 2 = improvement <25% vs. improvement 50–74%, 3 = improvement <25% vs. improvement \ge 75%, 4 = improvement 25–49% vs. improvement 50–74%, 5 = improve 25–49% vs. improvement \ge 75%, and 6 = improvement 50–74% vs. improvement \ge 75%. *p < 0.05, **p < 0.01, ***p < 0.001.

²Negative change scores indicate improvement.

with <PASI25. One-way analyses of variance were performed among these groups for each of the PRO measures. As can be seen by the size of the f-statistics, the DLQI was the most responsive of the PRO measures. In fact, only the DLQI was able to demonstrate statistically significant differences between responders and partial responders based on post-hoc significance tests among the four responder groups. These results for the DLQI total score with respect to differences among responder groups were similar to those reported previously in the literature, except that the improvement in DLQI total scores displayed in Table 7 was larger for each of the responder groups than for the equivalent responder groups described by Shikiar and colleagues in a study of efalizumab [28]. As was the case for the data displayed in Table 6, the responsiveness of the EQ-5D index and VAS scores were generally the same as for most of the SF-36 scores. Finally, both the SF-36 MCS and PCS scores were responsive, but the MCS was substantially more responsive, indicating that the impact of the disease was both physical and mental, with the latter perhaps being more prominent for this study population.

Estimates of Minimum Important Differences

There is no one best way to estimate the MID for a PRO measure [21,34]. Table 8 contains three different anchorbased methods for estimating the MID based on data from this study. MID-1 contains the estimate obtained from the scores from the "near-responders," shown as the PASI 25-PASI 49 group in Table 7; MID-2 contains the estimate corresponding to "partial responders" in the same table 7. MID-3 corresponds to the difference between non-responders for the PGA (defined as patients who had no change in score or a decrease in score by one point on this 7-point scale) and minimal responders for this same measure (defined as patients who improved by 1 or 2 points). The distribution-based estimates, the SEM and one-half the standard deviation of baseline scores are also reported in Table 8.

Estimates for the DLQI MID ranged from 4.05 (for MID-1) to 6.95 (for MID-2), while the SEM was 2.33 and one-half standard deviation was 3.59. The MID results for the SF-36 PCS ranged from 0.51 (for MID-3) to 3.91 (for MID-1), with the SEM estimated as 2.71 and one-half

Change in Measure	MID-1: PASI Improvement 25–49%	MID-2: PASI Improvement 50–74%	MID-3: Difference Between Non-Responders (n = 34) and Minimal Responders (n = 41) on PGA	SEM	0.5 SD
DLQI Total Score	-4.05 (4.95)	-6.95 (5.71)	-5.69	2.33	3.59
SF-36					
Physical Functioning ¹	N/A	N/A	N/A	N/A	N/A
Role-Physical	3.41 (51.35)	14.29 (31.20)	10.51	15.02	18.78
Bodily Pain	11.59 (22.96)	16.76 (22.74)	9.05	10.76	12.69
General Health	4.41 (18.37)	9.19 (11.29)	4.97	9.67	10.31
Vitality	4.77 (20.44)	6.90 (17.43)	6.54	8.22	11.40
Social Functioning	-1.70 (27.36)	13.10 (17.44)	13.62	10.67	13.78
Role-Emotional	-7.58 (38.40)	17.46 (34.35)	24.71	14.59	17.70
Mental Health	0.18 (18.66)	9.52 (13.53)	4.90	6.10	9.65
Physical Summary Score (PCS)	3.91 (10.88)	2.57 (7.78)	0.51	2.71	5.12
Mental Summary Score (MCS)	-1.82 (11.38)	6.05 (6.90)	6.61	3.89	5.61
EQ-5D					
Index Score	0.10 (0.24)	0.20 (0.21)	0.09	0.22	0.14
VAS-General Health	3.82 (22.07)	8.43 (11.24)	4.59	N/A	10.34

Table 8: Estimates of MCID for PRO measures

Note: MID-1 corresponds to the score for the PASI 25–49 group; MID-2 corresponds to the score for the PASI50-74; for MID-3 and MID-4, reliability estimates for computing SEM were obtained from the data in this study for the DLQI and from estimates found in the literature for the SF-36 and EQ-5D.

¹MID estimates are not provided for the SF-36 Physical Function domain since there were not significant differences among responder groups.

standard deviation estimates as 5.12. For the MCS, the MID estimates included a decrease of 1.82 points based on a PASI improvement of 25–49%, but the other two MIDs were 6.05 and 6.61, respectively. The SEM for the MCS was 3.89 and one-half standard deviation was 5.61. Consistent with the MCS findings, decreases were observed for the Role-Emotional and Social Functioning domains for the MID-1 definition. The differences between non-responders and minimal responders ranged from 4.90 for Mental Health to 24.71 for Social Functioning (Table 8). The results for the EQ-5D index score demonstrated an MID ranging from 0.09 (for MID-3) to 0.20 (for MID-2). For the EQ-5D VAS, the available estimates ranged from 3.82 (MID-1) to 8.43 (MID-3).

Discussion

A Phase II randomized clinical trial of adalimumab in moderate to severe plaque psoriasis provided the opportunity to evaluate the validity and responsiveness to clinical change of three PRO assessment instruments – one dermatology-specific instrument and two general health status instruments – all used as endpoints in the study. All analyses were performed on a blinded basis, since the main focus of these secondary analyses was on the psychometric qualities of the PRO instruments.

Although developed for a general population with dermatologic diseases, the DLQI has most frequently been applied to patients with plaque psoriasis [27]. More recently, the DLQI has been used as an endpoint in clinical trials involving the newer class of biologics for treat-

ment of moderate to severe psoriasis, including alefacept [6,7], etanercept [9,10], and efalizumab [8,11]. The present study further establishes the reliability and validity of the DLQI and its responsiveness to change in the clinical status of patients over the course of a 12-week clinical trial, confirming previous findings [28]. Changes in the DLQI total score demonstrated significant and sizeable correlations with independently obtained physicianassessed changes in the clinical statuses of patients. This indicates that the alleviation of psoriatic signs, as determined by clinical assessments, results in significant and marked improvement in dermatologic-related functional limitations and quality of life in patients with moderate to severe plaque psoriasis. Based on this study, the DLOI is a psychometrically sound and responsive measure of psoriasis-specific outcomes that captures more comprehensively the impact of clinical signs and symptoms on patient well-being.

Data were also used to derive estimates of the MID of the DLQI. Although the MID is defined as the smallest difference that a patient would perceive as beneficial, there were no patient-based assessments of change in this study. Hence, lacking a patient-based anchor, the data do not provide the basis for determining the smallest score that a patient would perceive as beneficial. We used both the PASI and the PGA, as well as two distributional approaches to derive estimates of the MID of the DLQI. These estimates ranged from 2.33–6.95. However, we believe that the PASI 50 is too conservative for estimating the minimum change that patients will find beneficial.

Therefore, we believe the estimate based on PASI improvement of 25-49% or between non-responders and minimal responders provide better estimates of MID. Therefore, the results indicate that the MID is in the range of approximately 2.3-5.7, which is slightly higher than the range of estimates derived from Shikiar et al. [28] in an analysis of two clinical trials involving another psoriasis therapy. The distributional approaches resulted in the lowest estimates of MID for the DLQI, but it should be noted that the distributional approach to estimating the MID is considered supportive of the anchor-based methods [22,35]. For example, the one-half standard deviation estimate is certainly clinically meaningful, but is likely not a minimum magnitude of change. Finally, the range of estimates incorporates another previous estimate of the MID of the DLQI of 5.0 [36].

Two general PRO measures were used in this study. In general, the EQ-5D index and VAS scores demonstrated higher correlations than the SF-36 scale scores with the clinical endpoints (Table 5). However, the responsiveness of these two EQ-5D scores was generally the same as the responsiveness of most of the SF-36 scores. Nonetheless, this study demonstrated that the EQ-5D performs at least as well as the SF-36 as a non-dermatologic specific PRO measure for this sample of moderate to severe psoriatic patients.

Although most of the SF-36 scores showed improvements associated with clinical outcomes, the MCS, Social Functioning, and Role-Emotional domain scores demonstrated decreases in the PASI 25-49% group. These findings may have been driven by several outliers and the relatively small sample size for this group. Alternatively, given that Bodily Pain and other physical domains may be more related to the signs and symptoms of psoriasis than Role-Limitations and Social Functioning, small improvements in PASI scores may not be directly associated with changes in these PRO domains. That is, larger changes in clinical outcomes may be needed to significantly impact the areas of physical function and well-being. This idea seems to be supported by the observed changes in the PASI 50-74% and other analyses. However, the SF-36 domain and summary scores demonstrated consistently reasonable validity and were correlated with clinical endpoints and DLQI scores.

The SF-36 PCS and MCS scores demonstrated good evidence of validity and responsiveness in this sample of patients with moderate to severe plaque psoriasis. There were demonstrable associations between changes in PASI score categories and changes in PCS scores, with the largest improvements seen in the PASI75 responder groups. The MID estimates for the PCS were in the range of 0.51– 3.91, with the best estimate at approximately 2.5 points. The SEM estimate (2.71) also supports this range of MID values for the PCS. These results are consistent with previous research on the PCS scores in rheumatoid arthritis and other chronic diseases [29, 37]. The MID findings for the MCS were somewhat weaker, but there is evidence that a change of 4–6 points is certainly clinically meaningful. The MID for the EQ-5D index score was in the range of 0.09–0.22.

Given the impact of psoriasis on the functional ability of patients the importance attached to assessing physical function in psoriasis patients, the results of the present study provide positive support for the use of a dermatology-specific health-related PRO measure, the DLQI, in the assessment of psoriasis and responses to treatment. In addition, the correlation of SF-36 and DLQI indicates that disease-related changes in the SF-36 are largely dependent on two specific domains, Bodily Pain and Social Functioning. It appears that the DLQI total score, as a single index score, adequately captures the functional and psychosocial impact of moderate to severe plaque psoriasis. Further, the DLQI does so in a way that is substantially more responsive than the general health-related quality of life measures used to assess changes in patients' underlying clinical statuses. The importance of the DLQI in measuring psoriasis patients' disease statuses, both at baseline and after treatment, is underlined by recent UK guidelines that recommend the DLQI serve both as an indicator of biologic therapy need and adequate treatment response [14].

There were several limitations to the present analysis. The first relates to sample size and selection. The sample was limited to those meeting the inclusion/exclusion criteria if this Phase II clinical trial. Since this was a Phase II study, the sample size was smaller (N = 147) than typical Phase III studies involving moderate to severe psoriasis, thereby requiring even one to use even greater caution in extrapolating the results of this analysis. Other applications of the PRO instruments (e.g., other clinical settings or settings including non-biologic treatments) might not involve the same exclusions. Therefore, generalizability of these results may not be applicable to all clinical settings. Second, the DLQI is not the only dermatology-specific instrument to assess the impact of psoriasis on physical function and psychosocial factors. Other instruments have been developed [38, 39], but have not been used as frequently as the DLQI in psoriasis trials. Nonetheless, results reported here do not indicate whether the DLQI has relative advantages or disadvantages to these instruments. Finally, given that the MID denotes the minimum change that a patient would find beneficial, anchoring the estimates of MID to patient assessments of severity or change would prove useful, and the current Phase II trial did not include such assessments.

Conclusion

The findings of this study highlight the importance of capturing PRO measures in clinical trials of moderate to severe plaque psoriasis. This analysis provides additional evidence supporting the psychometric qualities and responsiveness of the DLQI as a disease-specific measure of PROs in psoriasis. The DLQI MID was determined as ranging from 2.3-5.7 points. While the DLQI provides the most reliable measure of clinical change, the data from this study demonstrate that the SF-36 and EQ-5D performed well as general measures of health status outcomes. While the SF-36 has been used in previous studies comparing psoriasis treatments [6,7,30], to date, there have been few applications of the EQ-5D in clinical trials of patients with moderate to severe plaque psoriasis. The results of this study indicate that these two instruments should be considered as a general health outcome measure in future clinical trials.

Competing interests

The work reported here was performed under contract by the United BioSource Corporation to Abbott Laboratories. Authors 1, 4, and 5 are employees of the United Bio-Source Corporation, which performed this work under contract. Authors 2 and 3 are current employees of Abbott Laboratories.

Authors' contributions

RS was responsible for the planning and analysis of the research reported here. MKW provided key analysis and interpretation of the data, and revised the manuscript critically for important intellectual content. MMO contributed important clinical insights into the research. CST contributed suggestions for analysis and provided analytical support for the research. DAR contributed to the analvsis and interpretation of the research and reviews of the manuscript. All authors reviewed and approved this manuscript prior to submission.

Acknowledgements

The work reported here was performed under a contract to the United BioSource Corporation from Abbott Laboratories. Lisa E. Melilli, DrPH, formerly of Abbott Laboratories contributed substantial important scientific and analytical suggestions and insights, and reviews of the manuscript. Mary Cifaldi, PhD, RPh., MSHA, of Abbott Laboratories provided invaluable reviews and comments on previous drafts of this manuscript. The authors thank Michael Nissen, ELS, of Abbott Laboratories for his editing assistance in the development of this manuscript.

Portions of these results were presented at poster and podium sessions at the European Meeting of the International Society for Pharmacoeconomic Research in Florence, Italy, November 2005.

References

- de Arruda LH, De Moraes AP: The impact of psoriasis on quality of life. Br | Dermatol 2001, 144:33-36.
- 2 Finlay AY: Quality of life assessments in dermatology. Semin Cutan Med Surg 1998, 17:291-296.

- Rapp SR, Cottrell CA, Leary MR: Social coping strategies associ-3. ated with quality of life decrements among psoriasis patients. Br J Dermatol 2001, 145:610-616. Wahl A: The impact of psoriasis on psychosocial life domains.
- 4. A review. Scand | Caring Sci 1997, 11:243-249.
- 5. Weiss SC, Kimball AB, Liewehr DJ, Blauvelt A, Turner ML, Emanuel E]: Quantifying the harmful effect of psoriasis on healthrelated quality of life. J Am Acad Dermatol 2002, 47:512-518.
- Feldman SR, Menter A, Koo JY: Improved health-related quality 6. of life following a randomized controlled trial of alefacept treatment in patients with chronic plaque psoriasis. Br J Dermatol 2004, 150:317-326.
- 7. Finlay AY, Salek MS, Haney J: Intramuscular alefacept improves health-related quality of life in patients with chronic plaque psoriasis. Dermatology 2003, 206:307-315.
- Gordon KB, Papp KA, Hamilton TK, Walicke PA, Dummer W, Li N, 8. Bresnahan BW, Menter A: Efalizumab for patients with moderate to severe plaque psoriasis: a randomized controlled trial. JAMA 2003, 290:3073-3080.
- Gottlieb AB, Matheson RT, Lowe N, Krueger GG, Kang S, Goffe BS, 9. Gaspari AA, Ling M, Weinstein GD, Nayak A, Gordon KB, Zitnik R: A randomized trial of etanercept as monotherapy for psoriasis. Arch Dermatol 2003, 139:1627-32; discussion 1632.
- 10 Leonardi CL, Powers JL, Matheson RT, Goffe BS, Zitnik R, Wang A, Gottlieb AB: Etanercept as monotherapy in patients with psoriasis. N Engl J Med 2003, 349:2014-2022
- 11. Menter A, Kosinski M, Bresnahan BW, Papp KA, Ware JEJ: Impact of efalizumab on psoriasis-specific patient-reported outcomes. Results from three randomized, placebo-controlled clinical trials of moderate to severe plaque psoriasis. J Drugs Dermatol 2004, 3:27-38.
- 12. Kirby B, Richards HL, Woo P, Hindle E, Main CJ, Griffiths CE: Physical and psychologic measures are necessary to assess overall
- psoriasis severity. J Am Acad Dermatol 2001, 45:72-76. Krueger GG, Feldman SR, Camisa C, Duvic M, Elder JT, Gottlieb AB, 13 Koo J, Krueger JG, Lebwohl M, Lowe N, Menter A, Morison WL, Prystowsky JH, Shupack JL, Taylor JR, Weinstein GD, Barton TL, Rolstad T, Day RM: Two considerations for patients with psoriasis and their clinicians: what defines mild, moderate, and severe psoriasis? What constitutes a clinically significant improvement when treating psoriasis? J Am Acad Dermatol 2000, 43:281-285
- 14. Smith CH, Anstey AV, Barker JN, Burden AD, Chalmers RJ, Chandler D, Finlay AY, Grifitths CE, Jackson K, McHugh NJ, McKenna KE, Reynolds NJ, Ormerod AD: British Association of Dermatologists guidelines for use of biological interventions in psoriasis 2005. Br | Dermatol 2005, 153:486-497.
- 15. Finlay AY, Khan GK: Dermatology Life Quality Index (DLQI)-a simple practical measure for routine clinical use. Clin Exp Dermatol 1994, 19:210-216.
- Ware JE, Snow KK, Kosinski MK, Gandek B: SF-36 Health Survey: Manual and Interpretation Guide. Boston, The Health Institute, New England Medical Center; 1993.
- 17. EuroQol Group: EuroQol--a new facility for the measurement of health-related quality of life. Health Policy 1990, 16:199-208.
- 18. Kind P: The EuroQol Instrument: An index of health-related quality of life. In Quality of Life and Pharmacoeconomics in Clinical Trials 2nd edition. Edited by: Spilker B. Philadelphia, Lippincott-Raven Publishers; 1996:191-201.
- 19. Food and Drug Administration: Guidance for industry. Patientreported outcome measures: use in medical product development to support labeling claims. Draft.
- 20. Juniper EF, Guyatt GH, Willan A, Griffith LE: Determining a minimal important change in a disease-specific Quality of Life Questionnaire. | Clin Epidemiol 1994, 47:81-87.
- Gordon KB, Langley RG, Leonardi C, Toth D, Menter MA, Kang S, Heffernan M, Miller B, Hamlin R, Lim L, Zhong J, Hoffman R, Okun MM: Clinical response to adalimumab treatment in moderate to severe psoriasis patients: double-blind, randomized controlled trial and open-label extension study. JAAD in press.
- 22. Ashcroft DM, Wan Po AL, Williams HC, Griffiths CE: Clinical measures of disease severity and outcome in psoriasis: a critical appraisal of their quality. Br | Dermatol 1999, 141:185-191.
- 23. Fredriksson T, Pettersson U: Severe psoriasis--oral therapy with a new retinoid. Dermatologica 1978, 157:238-244.

- Badia X, Mascaro JM, Lozano R: Measuring health-related quality of life in patients with mild to moderate eczema and psoriasis: clinical validity, reliability and sensitivity to change of the DLQI. The Cavide Research Group. Br J Dermatol 1999, 141:698-702.
- Lewis V, Finlay AY: 10 years experience of the Dermatology Life Quality Index (DLQI). J Investig Dermatol Symp Proc 2004, 9:169-180.
- Shikiar R, Bresnahan BW, Stone SP, Thompson C, Koo J, Revicki DA: Validity and reliability of patient reported outcomes used in Psoriasis: results from two randomized clinical trials. *Health Qual Life Outcomes* 2003, 1:53.
- Ware JE, Kosinski MK, Keller SD: SF-36 Physical and Mental Health Summary Scales: A User's Manual. Boston, The Health Institute, New England Medical Center; 1994.
- Wahl A, Loge JH, Wiklund I, Hanestad BR: The burden of psoriasis: a study concerning health-related quality of life among Norwegian adult patients with psoriasis compared with general population norms. J Am Acad Dermatol 2000, 43:803-808.
 Wyrwich KW, Nienaber NA, Tierney WM, Wolinsky FD: Linking
- Wyrwich KW, Nienaber NA, Tierney WM, Wolinsky FD: Linking clinical relevance and statistical significance in evaluating intra-individual changes in health-related quality of life. Med Care 1999, 37:469-478.
- Wyrwich KW, Tierney WM, Wolinsky FD: Further evidence supporting an SEM-based criterion for identifying meaningful intra-individual changes in health-related quality of life. J Clin Epidemiol 1999, 52:861-873.
- Norman GR, Sloan JA, Wyrwich KW: Interpretation of changes in health-related quality of life: the remarkable universality of half a standard deviation. Med Care 2003, 41:582-592.
- 32. Beaton DE, Boers M, Wells GA: Many faces of the minimal clinically important difference (MCID): a literature review and directions for future research. Curr Opin Rheumatol 2002, 14:109-114.
- Hays RD, Farivar SS, Liu H: Approaches and recommendations for estimating minimally important differences for healthrelated quality of life measures. *Journal of Chronic Obstructive Pulmonary Disease* 2005, 2:63-67.
- Khilji FA, Gonzalez M, Finlay AY: Clinical meaning of change in Dermatology Life Quality Index. Br J Dermatol 2002, 147:50 (abstract).
- Chren MM, Lasek RJ, Quinn LM, Mostow EN, Zyzanski SJ: Skindex, a quality-of-life measure for patients with skin disease: reliability, validity, and responsiveness. J Invest Dermatol 1996, 107:707-713.
- McKenna SP, Lebwohl M, Kahler KN: Development of the US PSORIQoL: a psoriasis-specific measure of quality of life. Int J Dermatol 2005, 44:462-469.

